Author's response to reviews

Title: Planning and process evaluation of a multi-faceted influenza vaccination implementation strategy for health care workers in acute health care settings

Authors:

Josien Riphagen-Dalhuisen (j.riphagen@umcg.nl)
Gerard Frijstein (g.frijstein@amc.uva.nl)
Nannet van der Geest-Blankert (a.vanderGeest@amd.ru.nl)
Marita Danhof-Pont (m.b.danhof-pont@lumc.nl)
Herbert de Jager (h.dejager@erasmusmc.nl)
Nita Bos (n.bos@umcutrecht.nl)
Ed Smeets (eej.smeets@mumc.nl)
Marjan de Vries (m.j.t.de.vries@umcg.nl)
Pieter Gallee (pmm.gallee@dienst.vu.nl)
Eelko Hak (e.hak@rug.nl)

Version: 6 Date: 18 April 2013

Author's response to reviews: see over
Date
18 April 2013

Subject
Manuscript submission after third revision

Dear Professor Icardi, dear Mrs. Harris,

Please find enclosed the third revision of the article by J. Riphagen-Dalhuisen et al. entitled ‘Planning and process evaluation of a multi-faceted influenza vaccination implementation strategy for health care workers in acute health care settings’.

Following the minor comments of the reviewer, we have adjusted and adapted our manuscript again. We also asked a professional editing service (Whitehorse Communication and Design) to correct the quality of written English. We believe this has led to a greatly improved version of our manuscript.

Below you will find the point-by-point response to the reviewers’ comments.

We hope this manuscript is now acceptable for publication in *BMC Infectious Diseases*.

Yours faithfully,

On behalf of all authors,

J. Riphagen-Dalhuisen, MD, PhD research fellow

---

**Response**

1) Abstract P1 S2 "reads better to move "in all European countries" to end of sentence
   - we have changed this
2) Abstract P2 S1 - can drop "the process of" in methods
   - we have deleted this
3) Background - P1S2 - I would write "continue" rather than "continues" but maybe my grammar is wrong? just check with someone that does know to be sure
   - We have corrected this, and the entire manuscript is corrected by a professional editing service to improve the quality of written English.